Author's response to reviews

Title: Stratification of Coronary Artery Disease Patients for Revascularization Procedure Based on Estimating Adverse Effects

Authors:

Sebastian Pölsterl (sebastian.poelsterl@tum.de)
Maneesh Singh (maneesh.singh@sri.com)
Amin Katouzian (amin.katouzian@cs.tum.edu)
Nassir Navab (nassir.navab@tum.de)
Adnan Kastrati (kastrati@dhm.mhn.de)
Lance Ladic (lance.ladic@siemens.com)
Ali Kamen (ali.kamen@siemens.com)

Version: 5 Date: 5 January 2015

Author's response to reviews:

We renamed the section entitled 'Materials and Methods' into 'Methods' and removed inline highlights that was used to reference the reviewers previous comments.